Author's response to reviews

Title: What they fill in today, may not be useful tomorrow: Lessons learned from a study of medical records at the Women hospital in Tabriz, Iran

Authors:

Faramarz Pourasghar (Faramarz.pourasghar@ki.se)
Hossein Malekafzali (afzali@hbi.ir)
Alireza Kazemi (dr_alireza_kazemi@yahoo.com)
Johan Ellenius (Johan.Ellenius@ki.se)
Uno GH Fors (Uno.Fors@ki.se)

Version: 4 Date: 23 April 2008

Author's response to reviews:

Dear editor, we have now revised the manuscript according to the latest comments from you.

1- Author list: The affiliation section of the manuscript was revised completely. The authors have been identified by new numerical symbols.

2- For every affiliation, there is a country suffix.

3- The keyword section was removed from the abstract.

4- The section numbers were removed from the text.

5- The word “Introduction” was replaced by the word “Background”

6- The “objective” section was merged into the background section of the manuscript.

7- The initials were used for referring to the authors’ contribution (page 16).

8- Table has been reformatted and was added to the last page of the manuscript.